Author's response to reviews

Title: Effects of acupuncture treatment on depression insomnia: A study protocol of a multi-centre randomized controlled trial

Authors:

Yuan-Fang Chen (yfchen1@126.com)
Zhao-Hui Liang (liang_zhao_hui@163.com)
Neng-Gui Xu (ngxu8018@gzhtcm.edu.cn)
Zhen-Hua Xu (xzh197011@163.com)
Jian-Hua Liu (ljh1008@21cn.com)
Wen-Bin Fu (fuwenbin@139.com)
Shu-jun Xu (13826497503@163.com)

Version: 4 Date: 29 October 2012

Author's response to reviews: see over
1 About the language, we have been improving the style of written English.
2 About the inclusion and exclusion criteria, we used the inclusion criteria of insomnia DSMIV as the inclusion criteria.
3 Number 3 inclusion criterial, there is a spelling mistake and the right spelling is HAMD.
4 Number 4, we are still using PSQI instead of the ISI, because PSQI is the construction validity and reliability of this scale have been proven in previous studies and it has been validated for use in China.
5 Number 6, inclusion criterial is “Age between 16 years and 50 years or equal 16 years and 50 years”
6 The detail of statistic analyses has been described in the revised version of the article.
7 The method has been rewritten.
8 About the sample size calculation, we explained the how the primary outcomes (PSQI, and HAMD) related with the stated outcome used in the sample size calculation. And it is about the cure of depression insomnia, not depression or insomnia.